Author's response to reviews

Title: Mental health problems after divorce last longer in couples with high pre-divorce alcohol consumption: Results from the HUNT-study

Authors:

Kamilla Rognmo (kamilla.rognmo@fhi.no)
Fartein A Torvik (fartein.torvik@fhi.no)
Mariann Idstad ( mariann.idstad@fhi.no)
Kristian Tambs ( kristian.tambs@fhi.no)

Version: 2 Date: 8 April 2013

Author's response to reviews:

Dear Editor,

Enclosed please find our manuscript entitled “Mental health problems after divorce last longer in couples with high pre-divorce alcohol consumption: Results from the HUNT-study”, submitted for consideration for publication in BMC Public Health.

The manuscript has been altered in accordance with your comments. In the Background section of the abstract, the aim of the study was made clear. Also, a section, called Ethics, has been added to the Methods section, on page 7. In this section, it is explained that the study was approved by the Regional Committee for Medical and Health Research Ethics for Central Norway, that all respondents gave their written consent to participate in the study, and that the study was performed in accordance with the principles laid down in the Helsinki declaration.

The manuscript is original, not previously published and not under concurrent consideration elsewhere. All co-authors are in agreement regarding the content of the manuscript.

We thank you for considering the manuscript and letting us resubmit the revised version. We hope it will be of interest to BMC Public Health.

Best regards
Kamilla Rognmo
Norwegian Institute of Public Health
Division of Mental Health
P.O. Box 4404 Nydalen,
NO-0403 Oslo, Norway
Phone: +47 900 80 681, Fax: +47 223 53 605,
E-mail: kamilla.rognmo@fhi.no